Author's response to reviews

Title: Randomised controlled trial of a psychiatric consultation model for treatment of common mental disorder in the occupational health setting. Study Protocol.

Authors:

Christina M Van der Feltz-Cornelis (cfeltz@trimbos.nl)
Fransina J De Jong (sjong@trimbos.nl)
Rob Hoedeman (rob.hoedeman@arboned.nl)
Jolanda Meeuwissen (jmeeuwissen@trimbos.nl)
Iman Elfeddali (ielfeddali@trimbos.nl)

Version: 2 Date: 9 December 2006

Author's response to reviews: see over
Dear Editor,

Hereby I submit to you the following Article:

**Randomised controlled trial of a psychiatric consultation model for treatment of common mental disorder in the occupational health setting. Study Protocol.**

Authors:
Christina M van der Feltz-Cornelis, Fransina J de Jong, Rob Hoedeman, Jolanda Meeuwissen, Iman Elfeddali.

This is a study protocol paper of a trial that has been funded after a peer reviewed grant application process. The trial has been registered in the trialregister and been approved by the Medical Ethical Committee Zuid, Tilburg, the Netherlands. Medical Ethical Committee approval number is 5.127. As I understand, this means that this article can be published without additional peer review, therefore I will not suggest reviewers.

This article has not been published or submitted elsewhere.
I do hope that you will consider it for publication in your Journal, BMC Health Services Research, and remain,

Yours sincerely,

Christina M van der Feltz-Cornelis, psychiatrist-epidemiologist